Author's response to reviews

Title: Age and gender differences in disabling foot pain using different definitions of the Manchester Foot Pain and Disability Index

Authors:

Hylton B Menz (h.menz@latrobe.edu.au)
Tiffany K Gill (tiffany.gill@adelaide.edu.au)
Anne W Taylor (anne.taylor@adelaide.edu.au)
Catherine L Hill (catherine.hill@nwahs.sa.gov.au)

Version: 3 Date: 19 October 2011

Author's response to reviews:

All production editing requests have been addressed. In addition, to address the concern of referee 3, the following has been added to paragraph 3:

A key advantage of the MFPDI is that it purports to measure disabling foot pain, and may theoretically be used to identify a more severely affected subgroup of people with foot pain. From an epidemiological perspective, it is important that these two subgroups (i.e. non-disabling foot pain versus disabling foot pain) can be identified, as they are likely to have different risk factor profiles and very different foot health care needs.